Author's response to reviews

Title: Health related quality of life in cancer patients at the end of life, Translation, validation and longitudinal analysis of specific tools: study protocol for a randomized controlled trial

Authors:

Anne-Lise POIRIER (anne-lise.poirier@ico.unicancer.fr)  
Fabrice KWIATKOWSKI (Fabrice.KWIATKOWSKI@cip.fr)  
Jean-Marie COMMER (jean-marie.commer@ico.unicancer.fr)  
Bénédicte D'AILLIERES (benedicte.daillieres@ico.unicancer.fr)  
Virginie BERGER (virginie.berger@ico.unicancer.fr)  
Mariette MERCIER (mariette.mercier@univ-fcomte.fr)  
Franck BONNETAIN (Fbonnetain@cgfl.fr)

Version: 3 Date: 9 March 2012

Author's response to reviews:

To Editors-in-Chief of trials journal:
Doug Altman, Curt Furberg, Jeremy Grimshaw and Peter Rothwell

Subject: online submission of the following revised manuscript “Health related quality of life of cancer patients at the end of life: Translation, validation and longitudinal analysis of specific tools” (Ref 6605979846301417)

Please find enclosed our revised manuscript entitled: "Health related quality of life in cancer patients at the end of life: Translation, validation and longitudinal analysis of specific tools.” which we are pleased to submit to “Trials” journal for publication.

We extend our sincere thanks for accepting our paper pending minor revisions. We agree with these remarks and we have modified the title and the manuscript accordingly. You will find every modification in the text highlighted in yellow.

The Study was registered in March 2012 on ClinicalTrials.gov, registration number was NCT01545921.

All authors of this paper have read and approved the final version submitted.

All authors of this paper have no conflict of interest to disclose.

Thank you for considering our paper for publication in “trials” journal.

Yours sincerely,

Dr Anne Lise Poirier For and on behalf of all authors
Centre d’Evaluation Clinique en Oncologie,  
Centre Paul Papin  
2, rue Moll 49933 Angers cedex 9
&
EA 4184 Faculty of medicine, Burgundy University, Dijon France